Author's response to reviews

Title: Belt restraint reduction in nursing homes: design of a quasi-experimental study

Authors:
- Math JM Gulpers (m.gulpers@zw.unimaas.nl)
- Michel HC Bleijlevens (m.bleijlevens@zw.unimaas.nl)
- Erik van Rossum (erik.vanrossum@zw.unimaas.nl)
- Elizabeth Capezuti (ec65@nyu.edu)
- Jan PH Hamers (jph.hamers@zw.unimaas.nl)

Version: 5 Date: 18 February 2010

Author's response to reviews:

February 18th, 2010

Dear editor,

We are delighted that you considered our revised manuscript to be sufficient and accepted it for publication in BMC Geriatrics.

The editorial production team has completed their formatting checks of our article and requested us to check our manuscript for any small typographical changes we would like to make. We have made some minor typographical changes and we added some missing words in a revised version of our manuscript. All changes have been highlighted in the currently submitted revision.

Yours sincerely, on behalf of the co-authors

Math Gulpers